Author's response to reviews

Title: Characterisation of an aerosol exposure system to evaluate the genotoxicity of whole mainstream cigarette smoke using the in vitro gammaH2AX assay by High Content Screening

Authors:

Carolina Garcia-Canton (Carolina_Garcia-Canton@bat.com)
Graham Errington (Graham_Errington@bat.com)
Arturo Anadon (Anadon@vet.ucm.es)
Clive Meredith (Clive_Meredith@bat.com)

Version: 5 Date: 6 July 2014

Author's response to reviews: see over
Editorial Requests

1. Abstract
Please include the following subheadings in your Abstract as detailed in our Instructions for Authors (http://www.biomedcentral.com/bmcpharmacoltoxicol/authors/instructions/researcharticle#formatting-abstract):
Background, Methods, Results, Conclusions.
Please also remember to update the Abstract present on our submission system.

Subheadings now added to abstract both in the document and submission system

2. Title page
We would ask that you include each authors email address on the title page.

Emails included

3. Correct section headings
Please rename the following sections in order to comply with our journal formatting guidelines:
Introduction -> 'Background'
Materials and Methods -> 'Methods'
Discussion and Conclusions -> Two separate sections, 'Discussion' followed by 'Conclusions'

Headings renamed as indicated

4. Figures and tables
These should appear after the references.

Sections moved as requested